Author’s response to reviews

Title: The Effectiveness of Virtual Reality for People with a Mild Cognitive Impairment or Dementia: A Meta-analysis

Authors:

Oksoo Kim (ohong@ewha.ac.kr)

Yanghee Pang (anais0220@hanmail.net)

Jung-Hee Kim (jee90@catholic.ac.kr)

Version: 1 Date: 27 May 2019

Author’s response to reviews:

Thank you for your review comments. We incorporated your suggestions into the manuscript. Your feedback was extremely helpful in strengthening it.

We checked the comments and revised the manuscript and also edited it again according to the comments. Thank you for your invaluable time.

To make checking easier, we marked the revised parts in the manuscript in blue.

Detailed response table is in attach files.